Author’s response to reviews

Title: Motor and cognitive functioning in children treated for idiopathic clubfoot at the age of 3 years

Authors:

Julia Dillmann (Julia.Dillmann@psychol.uni-giessen.de)

Gudrun Schwarzer (gudrun.schwarzer@psychol.uni-giessen.de)

Christian-Dominik Peterlein (peterlei@med.uni-marburg.de)

Version: 2 Date: 07 Oct 2019

Author’s response to reviews:

-- Please ensure to assign your manuscript a section/category upon resubmission. E.g. General pediatric medicine and surgery.

We choose the section General Medicine and surgery.

-- Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

We now represent authors’ names using their full initials in the Authors’ Contributions section.

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We described now the role of the Funding in more detail.

-- If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

We did not use any abbreviations in the text.
-- Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

We removed it.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We uploaded a final clean Version.